Author’s response to reviews

Title: Assessing factors militating against the acceptance and successful implementation of cloud based health center from healthcare professionals' perspective: A survey of hospitals in Benue state northcentral Nigeria

Authors:

Patience Idoga (20153744@student.ciu.edu.tr)

MEHMET TOYCAN (mtoycan@ciu.edu.tr)

Halil Nadiri (hnadiri@ciu.edu.tr)

Erbuğ Çelebi (ecelebi@ciu.edu.tr)

Version: 6 Date: 30 Jan 2019

Author’s response to reviews:

Dear Editor,

We would like to thank for your positive response. We have included the following Figure caption inside the manuscript under the "Cloud based Health Center" section in Page 9, Line 219-220.

"Fig.2 CBHC system architecture (source: Figure created using the demo version of ConceptDraw Diagram software which adheres to the terms and conditions of ConceptDraw) [Here]"